Author's response to reviews

Title: Epidemiology of Community-Onset Staphylococcus aureus Infections in Pediatric Patients: An Experience at a Children's Hospital in Central Illinois

Authors:

Kanokporn Mongkolrattanothai (kmongkol@uic.edu)
Jean Aldag (JCA@uicomp.uic.edu)
Peggy Mankin (PEM@uicomp.uic.edu)
Barry Gray (bmg@uicomp.uic.edu)

Version: 3 Date: 21 June 2009

Author's response to reviews: see over
Author’s response to reviews

Title: Epidemiology of Community-Onset *Staphylococcus aureus* Infections in Pediatric Patients: An Experience at a Children’s Hospital in Central Illinois

Authors:

Kanokporn Mongkolrattanothai (kmongkol@uic.edu)

Jean C Aldag (JCA@uic.edu)

Peggy Mankin (PEM@uic.edu)

Barry M Gray (bmg@uic.edu)

Version 3

Date June 22, 2009

To the Editor

BioMed Central Infectious Diseases

Re: Submission of the revised manuscript

Dear Editor:

Thank you for your interest in considering our paper. We are grateful to the reviewers for the critical and constructive comments. Below is our response to the reviews

1) Statement suggested by a referee 2 was added to the discussion.

2) The authors' qualifications were removed from the Title Page.

3) The heading "Abstract" was added.

4) "Introduction" was changed to "Background" as suggested.

5) The manuscript was revised to conform to all points in manuscript formatting checklist.